**Author’s response to reviews**

**Title:** Identification, description and appraisal of generic PROMs for primary care: a systematic review

**Authors:**
Mairead Murphy (mairead.murphy@bristol.ac.uk)
Sandra Hollinghurst (sandra.bristolpapers@gmail.com)
Chris Salisbury (c.salisbury@bristol.ac.uk)

**Version:** 2 **Date:** 26 Oct 2017

**Author’s response to reviews:**

Dear Editor,

Thank you for accepting the paper subject to these small corrections. I have noted that the reviewers were satisfied with the changes, and did not require any further revisions. As per the editorial instruction, I have:

1. Removed the highlighted text which indicated where changes were made from the first version

2. Removed the cover letter

3. Changed the first heading of the abstract to “background”

4. Indicated that funders had no role in study design, analysis and writing the paper

5. Removed the funders from the acknowledgements
I have uploaded a new paper with these small changes. The additional files and figures remain unchanged.

Best regards,

Mairead Murphy